Author’s response to reviews

Title: Defining the gap: A systematic review of the difference in rates of diabetes-related foot complications in Aboriginal and Torres Strait Islander people and non Indigenous Australians

Authors:

Matthew West (maw835@newcastle.edu.au)

Vivienne Chuter (vivienne.chuter@newcastle.edu.au)

Shannon Munteanu (S.Munteanu@latrobe.edu.au)

Fiona Hawke (Fiona.Hawke@newcastle.edu.au)

Version: 4 Date: 12 Oct 2017

Author’s response to reviews:

Dear Editor,

Thank you for your response and the time you have taken to complete such a thorough review of the this article. All changes have been accepted and the 'clean version' has now been uploaded.

With my thanks and kind regards,

Matthew West